Author's response to reviews

Title: Effect of Flabellaria paniculata Cav. extracts on gastric ulcer in rats

Authors:

Margaret Sofidiya (toyin_sofidiya@yahoo.co.uk)
lilian C. Agufobi (chidychik2@yahoo.com)
Abedemi J. Akindele (ai_akindele@yahoo.com)
Johnson A. Olowe (olowei@yahoo.com)
Oluwole B. Familoni (famiolonio@yahoo.com)

Version: 3 Date: 9 September 2012

Author's response to reviews: see over
The Editor
BMC Complementary and Alternative Medicine
RE: MANUSCRIPT ID: 6983180836787564
Dear Sir,
I received your mail requesting an update on the submission of our revised manuscript “MS: 6983180836787564 - Effect of Flabellaria paniculata Cav. extracts on gastric ulcer in rats”.

As at today, I still could not assess the facility to carry out the histological examination of the rats’ stomach tissue as demanded by Prof. Zakaria, one of the reviewers. I have made frantic effort within my environment with no success. I understand the academic in my Institution that could help with the histological examination is currently on vacation.

Sir, I have therefore, included in the conclusion section of the manuscript that ‘detailed histological assessment will be part of the follow-up study on FPL, including the bioactivity-guided fractionation to determine the bioactive principle(s) in the plant’.

If this will not suffice, I will then solicit for more time to resolve the issue.

Thank you for your understanding and support.

Yours truly,

Dr. Margaret O. Sofidiya